Author’s response to reviews

Title: A child presenting with tuberculous spondylitis in a single cervical (C3) vertebra: a case report

Authors:

Manouri Senanayake Prof. (manouri.senanayake@gmail.com)
Irantha Karunaratne (iranthakarunaratne@hotmail.com)

Version: 4 Date: 28 May 2014

Author’s response to reviews:

Dear Editors

Thank you for the comments. We have made the corrections as recommended.

Prof M Senanayake